Author's response to reviews

Title: Focus Group Interviews examination of knowledge, attitude and practice to lead exposure in South Western Nigeria

Authors:

Eugenious O Adebamowo (e_adebamowo@yahoo.com)
Oluwole A Agbede (agbede@ui.edu.ng)
Clement A Adebamowo (cadebamo@yahoo.com)
Mynepalli K Sridhar (sridhar@ui.edu.ng)

Version: 3 Date: 9 January 2006

Author's response to reviews:

We have reviewed the paper as suggested.
The sampling frame has been described in more detail.
More detailed description of the method of data analysis has been given
We have discussed the FGD guide and the major content areas
The major topics have been included in Table 1
The findings and tables have been reorganized
We have included discussion about how the participants compared to the Nigerian population
The current regulatory environment in Nigeria has been discussed
The limitations section has been expanded and the impact of having such a heterogenous group discussed
Implications for future research has been included
Table 2 has been split
The title has been changed
The typographical and grammatical errors have corrected.